Author's response to reviews

Title: Socioeconomic inequalities in cause specific mortality among older people in France

Authors:

Gwenn Menvielle (Gwenn.Menvielle@inserm.fr)  
Annette Leclerc (Annette.Leclerc@inserm.fr)  
Jean-François Chastang (jean-francois.chastang@inserm.fr)  
Danièle Luce (Daniele.Luce@inserm.fr)

Version: 5 Date: 10 May 2010

Author's response to reviews:

Dear Editor,

I changed the manuscript following your requests.

Regarding the affiliations, I modified them to stick to your instructions. The specific organisation of research institutes in France explains why Inserm U1018 should be indicated first (this is the first part of the name of the Dept) and Inserm is indicated later (this is the name of the Institute).

Sincerely yours

Gwenn Menvielle